2nd Internation Users Meeting December 1st and 2nd, 1994

SECOND ANNOUNCEMENT

The meeting will be held at the Main Building of the International Centre for Theoretical Physics (ICTP, Strada Costiera 11 - Grignano, Trieste) on December 1st and 2nd, 1994.

November 30 th.

17.00/20.00 - Preliminary Registration

December 1 st

08.00/09.00 Registration 09.00/09.15 L.Fonda, Sincrotrone Trieste Introduction 09.15/09.30 G.Viani, Sincrotrone Trieste Current Status of Sincrotrone Trieste Company

Chairperson : L. Fonda

09.30/10.00 C.Rizzuto, INFM-Genova, Synchrotron Radiation: Development and Objectives in Europe 10.00/10.20 A.Wrulich, Sincrotrone Trieste, General Status of ELETTRA: Present-Day Performance and Outlook 10.20/10.40 - Coffee break

Chairperson : A.Savoia

10.40/11.05 R.Rosei, Sincrotrone Trieste, Presentation of the Beamlines and Research Programs 11.05/12.15 First Results of the Operating Beamlines

SuperEsca, G. Paolucci, Sincrotrone Trieste Esca Microscopy, M.Kiskinova, Sincrotrone Trieste Vuv Photoemission, P. Perfetti, CNR-Roma X-Ray Diffraction, M.Colapietro, CNR-Roma 12.45/14.30 Greetings by the Authorities at Trieste City Hall and Buffet Lunch 15.00/16.30 Visit to ELETTRA Laboratory 16.30/17.00 Poster Session 20.00/--.-- Conference dinner at Hotel Riviera

December 2 nd

08.00/09.00 Registration

Chairperson: G. Viani 09.00/09.20 M. Sancin, Area di Ricerca-Trieste, Presentation to the Users of the Area di Ricerca Services 09.20/09.50 G. Margaritondo, Sincrotrone Trieste Program Advisory Committee, Presentation of the Sincrotrone Trieste Users Policy 09.50/10.20 Discussion 10.20/10.40 Coffee Break

Chairperson: R.Rosei 10.40/11.10 D. Lynch, Iowa State University, Current Status of the Synchrotron Radiation Center and its Research Program 11.10/11.30 R. Wechsung, Microparts-Dortmund, LIGA Technology and its Applications 11.30/11.45 P. Giribona, Ital-TBS-Trieste, Applied Research and Development of Devices Activity of Ital-TBS on ELETTRA 11.45/12.00 B. Gasperetti, Sincrotrone Trieste, MicroMore Program at Elettra

Chairperson: G. Chiarotti 12.00/13.30 Research Institutions’ Representatives

Round Table: Programs of Research Institutions in relation to ELETTRA and Financing Possibilities 13.30/14.30 Lunch

Chairperson: P. Perfetti 14.30/17.00 Users’ General Assembly, convened by the Representatives 17.00 - End of the Session